Author's response to reviews

**Title:** How does lean work in emergency care? A case study of a lean-inspired intervention at the Astrid Lindgren Children's Hospital, Stockholm, Sweden

**Authors:**

Pamela Mazzocato (pamela.mazzocato@ki.se)  
Richard J Holden (richard.holden@vanderbilt.edu)  
Mats Brommels (mats.brommels@ki.se)  
Håkan Aronsson (hakan.aronsson@liu.se)  
Ulrika Bäckman (ulrika.backman@karolinska.se)  
Mattias Elg (mattias.elg@liu.se)  
Johan Thor (johan.thor@ki.se)

**Version:** 11  **Date:** 27 January 2012

Author's response to reviews:

Below we describe the changes made.

Tables: we are unable to display shading and colours within the tables. Can we therefore ask you to use another method of highlighting the shaded text e.g. by making it bold. All visible vertical lines should also be removed, and please note that we are unable to correctly display merged cells where the merged cell crosses rows: please re-layout your table without these merged elements Please place all tables in a separate section at the end of the manuscript. The table title should appear above the table and should be only one sentence long. Please place any other text as a legend (or key), which should appear below the table

- We changed tables accordingly

References: reference 43 appears to be a duplicate of reference 37. Please only reference each source once and amend the list and main text accordingly

- We have kept only one reference

Title: please use sentence case in the title, both in the manuscript file and the submission system. Capitalise only the first word and proper nouns, and do not include full stops/periods. For example, "The Title of My Prestigious Publication: A Scientific Study." should be changed to "The title of my prestigious publication: a scientific study"

- We have modified the title accordingly. We kept the capital letters on the name of the hospital: How does lean work in emergency care? A case study of a lean-inspired intervention at the Astrid Lindgren Children's Hospital, Stockholm, Sweden

Abstract: Please change the 'Findings' sub-heading to 'Results'
- We changed the text accordingly

Please remove figure placement text

- We have removed the text from the manuscript main text

Please place funding information in an "Acknowledgements" section

- We have moved that to acknowledgements

Figures: It is important for the final layout of the manuscript that the figures are cropped as closely as possible to minimise white space around the image. Our online figure guide contains full details for preparing files for submission and can be viewed here: http://www.biomedcentral.com/ifora/figures

- We have minimised the white space around the figures

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance

- We have checked the text